Author's response to reviews

Title: Oral cancer awareness of undergraduate medical and dental students

Authors:

Lachlan M Carter (carter.lachlan@virgin.net)
Graham R Ogden (g.r.ogden@dundee.ac.uk)

Version: 5 Date: 14 November 2007

Author's response to reviews: see over
Dear editor,

RE: MS 8193478814207154
Title: Oral cancer awareness of undergraduate medical and dental students
Lachlan M Carter and Graham R Ogden

Many thanks for accepting the above paper for publication in BMC Medical Education. The formatting changes have been completed on the manuscript and on the submission system.

Yours sincerely

Lachlan Carter          Graham Ogden